\nocite{karnes2004,karnes2005,dhhangul,HongYP91,Choi99}

\ifEUCmode
\bibliographystyle{halpha}
\bibliography{texbook1,texbook3,kotexguide-k}
\else
\bibliographystyle{apacite}
\bibliography{texbook1,texbook3,kotexguide}
\fi

